Author's response to reviews

Title: A Frailty Instrument for primary care: findings from the Survey of Health, Ageing and Retirement in Europe (SHARE)

Authors:

Roman Romero-Ortuno (romeror@tcd.ie)
Cathal D Walsh (cathal.walsh@tcd.ie)
Brian A Lawlor (lawlorba@tcd.ie)
Rose Anne Kenny (rkenny@tcd.ie)

Version: 4 Date: 19 August 2010

Author's response to reviews:

Dear Editorial team,

Please find attached our manuscript with the further formatting changes requested.

We have provided a functioning web link to the website for References 28 and 29.

We have removed shading within the tables and used bold to highlight text.

We have removed the visible vertical lines from the tables.

References 19, 21, 24 and 31 now include the full list of authors (all references have fewer than 30 authors).

Best regards,

The Authors.